Author's response to reviews

Title: Outcomes and factors influencing the survival in cases with spontaneous rupture of hepatocellular carcinoma: a multicenter study

Authors:

Hiroyuki Kirikoshi (hkirikos@med.yokohama-cu.ac.jp)
Satoru Saito (ssai1423@yokohama-cu.ac.jp)
Masato Yoneda (yoneda@med.yokohama-cu.ac.jp)
Koji Fujita (kfujita@yokohama-cu.ac.jp)
Hironori Mawatari (mawatari-ykh@umin.ac.jp)
Takashi Uchiyama (maruru0203@yahoo.co.jp)
Takuma Higurashi (takumahigurashi@yahoo.co.jp)
Kento Imao (kento_impact@yahoo.co.jp)
Takashi Sakaguchi (sakagt@ii.em-net.ne.jp)
Kazuhiro Atsukawa (accho@bf6.so-net.ne.jp)
Aya Sawabe (aya.sawabe@hotmail.co.jp)
Akira Kanesaki (kanesak-home@ezweb.ne.jp)
Hirokazu Takahashi (hirokazu@med.yokohama-cu.ac.jp)
Yasunobu Abe (a0121@yokohama-cu.ac.jp)
Masahiko Inamori (inamorim@med.yokohama-cu.ac.jp)
Noritoshi Kobayashi (norikoba@yokohama-cu.ac.jp)
Kensuke Kubota (kubotak@yokohama-cu.ac.jp)
Norio Ueno (nu1952@med.yokohama-cu.ac.jp)
Atsushi Nakajima (nakajima-tky@umin.ac.jp)

Version: 2 Date: 14 November 2008

Author's response to reviews:

Editorial Office
BMC Gastroenterology
November 14, 2008

Dear Editors,

We are sorry about the style of our previously submitted manuscript in detail entitled “Outcomes and factors influencing the survival in cases with spontaneous rupture of hepatocellular carcinoma: a multicenter study” (Manuscript ID: 1652799316235997) in BMC Gastroenterology.

We have changed the style of this manuscript consistent with that of BMC Gastroenterology and would like to re-submit in your journal. We are very thankful for your kindness.

Sincerely yours,
Hiroyuki Kirikoshi, M.D., Ph.D.
Gastroenterology Division
Yokohama City University Graduate School of Medicine
3-9 Fuku-ura, Kanazawa-ku, Yokohama 236-0004, Japan
E-mail: hkirikos@med.yokohama-cu.ac.jp
Phone: +81-45-787-2640, FAX: +81-45-784-3546